Author's response to reviews

Title: Even the best laid plans: a qualitative study of the experience of transitioning to wheelchair use among older adults and their care providers

Authors:

Edward M Giesbrecht (Ed.Giesbrecht@med.umanitoba.ca)
William C Miller (bill.miller@ubc.ca)
Roberta L Woodgate (Roberta.Woodgate@umanitoba.ca)

Version: 5 Date: 1 June 2015

Author's response to reviews:

Thank you for your response to our manuscript re-submission. We have addressed the two concerns raised as follows:

1. Informed consent was written. This has been added to Line 122 on page 5.
2. The Figure legend has been added to the manuscript after the References on Lines 665-667, page 30.